Author’s response to reviews

Title: Repeated False Reactive ADVIA Centaur® and Bio-Rad GeeniusTM HIV Tests in a Patient Self-Administering Anabolic Steroids

Authors:

Polina Tsybina (polly.tsybina@usask.ca)
Maurice Hennink (Maurice.Hennink@saskhealthauthority.ca)
Tania Diener (tania.diener@saskhealthauthority.ca)
Jessica Minion (jessica.minion@saskhealthauthority.ca)
Amanda Lang (amanda.lang@saskhealthauthority.ca)
Stephanie Lavoie (stephanie.s.lavoie@phac-aspc.gc.ca)
John Kim (john.kim@canada.ca)
Alexander Wong (awong37@gmail.com)

Version: 1 Date: 10 Dec 2019

Author’s response to reviews:

Reviewer 1:

1. Re: specific bands that were reactive on Geenius assay: thank you for the comment, it does seem useful to include this information. We added the specific bands that were present on Geenius assay when patient's sample was tested (page 5, line 105). We additionally incorporated this data in the discussion (page 9, paragraph starting with line 185).

2. Re: paragraph starting on line 172: this was rewritten (now on page 8, starting on line 174).